Author's response to reviews

Title: Mental health problems in the 10th grade and non-completion of upper secondary school: the mediating role of grades in a population-based longitudinal study

Authors:

Åse Sagatun (aase.sagatun@r-bup.no)
Sonja Heyerdahl (sonja.heyerdahl@r-bup.no)
Tore Wentzel-Larsen (tore.wentzel-larsen@r-bup.no)
Lars Lien (lars.lien@medisin.uio.no)

Version: 2  Date: 31 July 2013

Author's response to reviews:

Dear Editor

Thank you for your quick response.

I have now included information on where we have resaved permissions to use the data. This is done under “Study population and procedure” in the “Methods” on page 5, line 12:

“All parents received written information, and the students signed an informed consent form at baseline indicating their acceptance of the linkage of survey data to registry data. Permission to use the survey data was given by The Norwegian Institute of Public Health, the study was approved by the Regional Committee for Medical and Health Research Ethics and by the Norwegian Data Inspectorate.”

Yours sincerely,

Åse Sagatun, PhD
Corresponding author